Author's response to reviews

Title: Longitudinal study of leptin levels in chronic hemodialysis patients

Authors:

Ilia Beberashvili (iliab@asaf.health.gov.il)
Inna Sinuani (sinuanii@asaf.health.gov.il)
Ada Azar (liada@bezegint.net)
Hila Yasur (iliabeber@yahoo.com)
Leonid Feldman (leonidf@asaf.health.gov.il)
Zhan Averbukh (AVERBUKHZH@asaf.health.gov.il)
Joshua Weissgarten (weissgarten@asaf.health.gov.il)

Version: 5 Date: 8 June 2011

Author's response to reviews:

To: Nehme Gabriel, M.D.
Editor-in-Chief, Nutrition Journal.

Subject: Revision - Nutrition Journal -1355444765525489.

Dear Editor,

Thank you for your letter concerning our manuscript # 1355444765525489 entitled “Longitudinal study of leptin levels in chronic hemodialysis patients”.

Now we enclose clean version of the revised manuscript.

In the author list we removed qualifications of authors, as requested.
We removed all highlighting and tracking from the manuscript, as requested.
We removed the titles from the image files, as requested.
In figure 3 we changed P value to 0.61 (instead 0.606).
No other corrections were made in the manuscript.

Sincerely yours,
Ilia Beberashvili, MD.
Nephrology Division, Assaf Harofeh Medical Center